



Gansu Survey of Children and Families Papers

Gansu Survey of Children and Families

July 2006

Parental Education and Investment in Children's Human Capital in Rural China

Philip H. Brown Colby College

Follow this and additional works at: http://repository.upenn.edu/gansu_papers

Brown, Philip H., "Parental Education and Investment in Children's Human Capital in Rural China" (2006). *Gansu Survey of Children and Families Papers*. 8. http://repository.upenn.edu/gansu_papers/8

© 2006 by The University of Chicago. All rights reserved. Reprinted from *Economic Development & Cultural Change*, Volume 54, Issue 4, July 2006, pages 759-789. Publisher URL: http://www.journals.uchicago.edu/loi/edcc

This paper is posted at ScholarlyCommons. http://repository.upenn.edu/gansu_papers/8 For more information, please contact libraryrepository@pobox.upenn.edu.

Parental Education and Investment in Children's Human Capital in Rural China

Abstract

The landmark study of race and education in the United States known as the "Coleman Report" (Coleman et al. 1966) concluded that family characteristics are more important determinants of educational achievement than school quality or teacher experience, particularly in the early stages of schooling. From this result sprang two prominent lines of academic inquiry. The first focuses on so-called education production functions (e.g., Hanushek 1997), with an eye toward cost-benefit analyses of various investments in teachers and schools. These studies often pay little attention to family background variables, treating them as exogenous controls. The second line of inquiry seeks to promote social policies that foster student achievement by studying why family background has such a pronounced effect on children's acquisition of human capital.

Comments

© 2006 by The University of Chicago. All rights reserved. Reprinted from *Economic Development & Cultural Change*, Volume 54, Issue 4, July 2006, pages 759-789. Publisher URL: http://www.journals.uchicago.edu/loi/edcc

Parental Education and Investment in Children's Human Capital in Rural China

PHILIP H. BROWN Colby College

I. Introduction

The landmark study of race and education in the United States known as the "Coleman Report" (Coleman et al. 1966) concluded that family characteristics are more important determinants of educational achievement than school quality or teacher experience, particularly in the early stages of schooling. From this result sprang two prominent lines of academic inquiry. The first focuses on so-called education production functions (e.g., Hanushek 1997), with an eye toward cost-benefit analyses of various investments in teachers and schools. These studies often pay little attention to family background variables, treating them as exogenous controls. The second line of inquiry seeks to promote social policies that foster student achievement by studying why family background has such a pronounced effect on children's acquisition of human capital.

Among the latter group, Behrman et al. (1997) find that rural Pakistani children whose fathers completed junior secondary school score 31% higher on reading tests and 29% higher on mathematics tests than children whose fathers did not. Similarly, Case and Deaton (1999) show that the head of household's education influences literacy and numeracy scores for black South African high school students. These relationships have generally been found to be robust to the inclusion of various household, school, and communitylevel characteristics, suggesting that parental education has a real effect on children's human capital acquisition (Strauss and Thomas 1995). In addition, Glewwe and Jacoby (1994) find a strong relationship between mother's education and mathematics and reading test scores using matched householdschool data from Ghana. The fact that father's education does not also strongly

I am grateful to the Spencer Foundation, the Mellon Foundation, and the Center for Chinese Studies at the University of Michigan for supporting the fieldwork and writing of this research. Albert Park, David Lam, Jan Svejnar, and Bob Willis provided many helpful suggestions. This article also benefited tremendously from comments by Axel Anderson, Andrew Coleman, Eric Edmonds, Emily Hannum, Sandy Hofferth, Chris Ryan, Steve Salant, and two anonymous referees. Remaining errors are my own. Contact the author at phbrown@colby.edu.

© 2006 by The University of Chicago. All rights reserved. 0013-0079/2006/5404-0007\$10.00

influence child learning in this study suggests that human capital investments may depend on both parent and child gender, a finding consistent with the conclusions of Lillard and Willis (1994) for Malaysia and Thomas (1994) for Brazil, Ghana, and the United States.

Still, the reasons that parental education has such a pronounced effect on child learning are not very well understood. One explanation that is often posited is that more educated parents make greater investments in children's human capital (Strauss and Thomas 1995) by providing higher levels of goods and services that complement learning and by devoting more time to their children.¹ However, more educated parents in poor households without access to credit may face a trade-off between providing more goods and allocating more time to interacting with their children. Specifically, more educated parents may receive higher wages and thus may have a higher opportunity cost of time spent outside the workplace. Such parents may forgo some time spent interacting with children to spend more time working and may make greater investments in goods for their children's human capital production as a result. Alternatively, parental education may increase the efficiency or effectiveness of the time spent interacting with children, and more educated parents may thus forgo some time spent working in order to make greater time investments in their children's human capital. However, if the returns to education are higher for the children of more educated parents or if parental education positively influences parental preferences for children's education, then more educated parents may make greater investments in both goods and time, even in poor households. Unfortunately, in the absence of very restrictive assumptions about the functional form of the human capital production function and about the degree of substitutability of goods and time in human capital production, theory has little to say about the effect of parental education on investments in children's human capital.²

Whether and how more educated parents make greater investments in their children's human capital are thus important empirical questions. Due to data limitations, however, relatively few empirical studies have been undertaken. Sathar and Lloyd (1994) investigate the impact of parental education on educational expenditures using survey data from Pakistan. They find that household spending on children's education is up to 75% higher if mothers ever attended school relative to households wherein mothers did not. Similarly,

¹ Hereafter, the term "goods" will refer both to goods and to services used in human capital production.

² Comparative static results showing ambiguous effects of parental education on investments for simple models featuring Cobb-Douglas human capital production and quasilinear human capital production are available from the author on request.

Behrman et al. (1999) analyze how mother's education affects parental time allocation using household data from India. Controlling for workforce participation, they find that literate mothers spend more time than illiterate mothers on total time allocated to "home care," which is defined as caring for children and performing household chores. These results suggest that more educated parents may make greater investments in both goods and time, although neither article addresses this question specifically.

Using a survey of children and their families, teachers, and school and village leaders in 100 rural villages in Gansu province, China, this study examines how parental education affects educational investments in children, focusing on investments in goods and time. The detailed data used in this work afford several innovations. First, the data capture more complete measures of educational investments than those used elsewhere in the literature. For example, nonrequired education-related expenditures, whether the household owns children's reading materials, and whether the household has a designated study area for the sampled child measure investments in goods used in children's human capital production. The number of hours that parents spend helping their children with homework each week, whether parents read to the sampled child, and whether parents discuss their children's school performance with teachers measure parental investments in time used in children's human capital production. Second, a comprehensive teacher ranking captures numerous aspects of teacher quality that are generally unavailable elsewhere. Moreover, teachers in Gansu generally follow a single cohort through primary school, suggesting that the teacher ranking is likely to measure both past and present teacher quality. Third, the relatively large number of villages allows for village fixed effects to control for school quality, local socioeconomic conditions, labor market opportunities, and other forms of unobserved heterogeneity at the local level. Finally, a child cognitive development measure is employed to control both for innate ability and expected returns to child education, a key omitted variable in most studies.

It is found that more educated parents allocate higher levels of both goods and time to their children's human capital production, even controlling for wealth, teacher quality, village fixed effects, and child cognitive development. Evidence suggests that more educated parents expect higher returns to education for their children, offering one reason why parents in resource-constrained households make greater investments in both goods and time. It is also found that the effect of mother's and father's education differs, with a marginal year of mother's education having a larger impact on time investments than a marginal year of father's education.

The remainder of this article is organized as follows: Section II provides an

overview of China's rural education system; Section III describes the empirical strategy and discusses some identification issues; Section IV introduces the data and variables; the effect of parental education on investments in children's human capital is analyzed empirically in Section V; and Section VI concludes.

II. China's Rural Educational System

Education is widely seen as a key determinant of future income and occupational attainment in modern China (Davis 1992; Knight and Song 1993; Hannum 1998). At a minimum, education helps rural residents take advantage of new farming techniques and improves access to off-farm jobs that require literacy or numeracy (Unger 2002). Furthermore, for people who are able to attain high levels of education, legally sanctioned urban residency and nonmanual work become possible. That is, students are granted urban residency permits (*hukou*) upon admission to an urban specialized secondary school (*zhong zhuan*) or university (Wu and Treiman 2004). These specialized secondary schools and universities are open to all students on the basis of merit, at least in principle, providing a strong incentive for rural children to pursue education.

However, economic decentralization in the post-Mao period left responsibility for funding education to the county and village governments (Hannum 1998), and in recent years, tight budgets at the local level have shifted the burden to parents through a series of school fees (Unger 2002). Schooling is often relatively expensive as a result, with school fees representing a sizable portion of rural household's incomes. Paying for schooling thus exacerbates economic hardships in poor rural villages (Tsang 1994).

Although schooling is generally underfunded, the central government made 9 years of education compulsory in the late 1990s. Nearly all Chinese children thus enroll in primary school (Unger 2002; Connelly and Zheng 2003).³ Because of the high cost, however, resource-constrained parents often withdraw their children before the 9 compulsory years of education have been completed, with academic performance and child gender strongly influencing this decision. For example, Brown and Park (2002) show that academically weak girls typically drop out of primary school (*xiao xue*) while academically weak boys generally finish primary school and enroll in junior secondary school (*chu zhong*). This finding is consistent with those of Hannum and Xie (1994), who describe sons receiving a disproportionate share of education in resource-constrained households because virilocal marriage (in which daughters marry out of the

³ Unger (2002, 185) writes, "To a degree that is at once touching and inspiring, parents in these poor villages generally make major sacrifices to keep their children in school, even when it means cutting back on the quality or quantity of food consumed or on other very basic essentials."

family) is the dominant practice in China and because sons are generally responsible for caring for parents in their old age.

While children are still in school, rural parents often make significant nonrequired investments in their children's education (Hannum 2005) to increase their chances for gaining off-farm employment and admission to competitive secondary schools and universities. Such investments may include the purchase of educational goods and services (e.g., dictionaries, notebooks, reading material, desks, and tutors) and the provision of time for educational purposes (e.g., time spent helping children with homework, time spent reading to children, and time spent discussing a child's academic performance with his or her teachers) as well as other forms (Kong 2003).

Teacher and school quality vary widely across and within provinces. Despite high enrollment fees, school budgets in rural parts of China are often inadequate to pay teachers on time, to purchase supplies, or to make needed structural repairs. Classrooms may lack basic infrastructure such as a sufficient number of chairs and desks, functional blackboards, and electricity.⁴ Top graduates from teachers' colleges are typically assigned to key schools in urban areas, leaving rural areas with teachers who may be less motivated or less well trained (Sargent and Hannum 2005) when they are available at all. Many rural communities cope with teacher shortages by hiring community members who have had no formal training to teach classes (Wang 2002). Furthermore, teachers in much of rural China follow student cohorts through school; students who attend school in poorly equipped classrooms or who are assigned to low quality teachers face high hurdles in their studies.

All but the largest rural villages have a single primary school comprising 6 years of education. Junior secondary schools consist of 3 years of schooling and are typically located in a larger village or in the township to which several villages belong. Due to the restrictive residency permit system, children almost always attend the nearest school; private schools and school choice are virtually nonexistent in most of rural China.⁵ Moreover, the vast majority of the children of rural-urban labor migrants remain in their home villages for schooling—even when their parents migrate for extended periods—because rural residents face much higher pricing in schooling, health care, and other services in urban

⁴ The worst schools I visited during survey fieldwork were barely functional: windows were missing or broken; a single piece of charcoal was used to heat a large classroom for an entire day; studentteacher ratios were very high due to inadequate funding for teacher salaries; and some teachers had acquired just a few years more education than their students. By contrast, schools in urban centers were generally modern, warm, and efficient.

⁵ Similarly, Li and Zhang (2004) state that China's residency permit system rules out endogenous sorting in the context of fertility.

areas (Solinger 1999; Du, Park, and Wang 2005). Finally, children may enroll in school as early as age 6, although many rural families delay school enrollment until age 7 or 8.

III. Empirical Specification

Investments in children's human capital (INVEST) are manifest in the provision of goods for educational purposes and time spent interacting with children. Investment in goods is measured by the household's total expenditure on nonrequired educational goods, that is, spending on school supplies and private tutoring. This measure excludes school fees, required textbooks, required uniforms, and other required spending for enrolled children. Investment in goods used in human capital production is also measured by whether the household has children's reading materials and whether the household has a designated study area for the sampled child. Parental time used in the production of children's human capital is measured by the total number of hours parents spend helping their children with homework each week and by measures for whether parents read to the sampled child and whether either parent meets with the sampled child's teachers to discuss his or her schooling.

The determinants of educational investment i in household h may be estimated by ordinary least squares or a linear probability model as follows:

$$INVEST_{ib} = \alpha + \beta_1 FED_{ib} + \beta_2 MED_{ib} + F_{ib}\beta_3 + K_{ib}\beta_4$$
$$+ \beta_5 (FED_{ib} \times MALE_{ib}) + \beta_6 (MED_{ib} \times MALE_{ib}) + e_{ib}, \quad (1)$$

where FED and MED are the number of grades completed by fathers and mothers, respectively; **F** is a vector of family characteristics including father's age (FAGE), mother's age (MAGE),⁶ the number of other enrolled children (ENROLLED), and the number of nonenrolled children (NONENR) in the household; **K** is a vector of child-specific characteristics such as sex (MALE), age, and grade level; and e_{ib} is an error term. Because father's and mother's education may affect investments in sons and daughters differently (see, e.g., Glewwe and Jacoby 1994; Lillard and Willis 1994; Thomas 1994), the sex of the sampled child is interacted separately with mother's and father's education.

Parental education has many correlates that may influence investments in children's human capital, and controlling for these effects may better identify

⁶ The timing of parental education may have implications for interpreting estimates because the quality and content of education have likely evolved over time, and controlling for measures such as parental age may help to mitigate this variation. A series of parental age dummies was also used to capture these effects, although the results are largely the same.

the true effect of parental education. For example, more educated parents may face less severe resource constraints; failing to control for household wealth may thus bias the estimated effect of parental education upward. Similarly, parental education may influence the decision to invest in children's human capital in response to either high or low teacher quality and school quality. A child's cognitive development may also affect the optimal household allocation because more educated parents may invest more in very gifted children or may help less gifted children by providing greater investments in their education. Additionally, local socioeconomic conditions, local labor market opportunities, and community preferences may also influence investment decisions. To test the robustness of the relationship between parental education and educational investments, wealth, teacher quality (TCHRRANK) and school and community characteristics (implemented via fixed effects), and a dummy for high cognitive development (HICOG) are added iteratively to the model described in equation (1). Finally, a model estimating the effect of parental education on educational investment i in household b in village vthat incorporates all of these controls is estimated as follows:

$$INVEST_{ibv} = \alpha + \beta_{1}FED_{ibv} + \beta_{2}MED_{ibv} + F_{ibv}\beta_{3} + K_{ibv}\beta_{4}$$

+ $\beta_{5}(FED_{ibv} \times MALE_{ibv}) + \beta_{6}(MED_{ibv} \times MALE_{ibv})$
+ $\beta_{7}WEALTH_{ibv} + \beta_{8}TCHRRANK_{ibv} + \beta_{9}HICOG_{ibv}$
+ $\beta_{10}(HICOG_{ibv} \times MALE) + \beta_{11}(FED \times HICOG_{ibv})$ (2)
+ $\beta_{12}(MED \times HICOG_{ibv})$
+ $\beta_{13}(FED \times HICOG_{ibv} \times MALE)$
+ $\beta_{14}(MED \times HICOG_{ibv} \times MALE) + \gamma_{iv} + e_{ibv},$

where γ_{ν} is a village fixed effect. Note that the cognitive development measure is interacted with parental education and child gender to allow for the possibility that parents invest differently in high-scoring boys and girls. If parental education is shown to have a positive effect on education investments even conditioning on this rich set of controls, then it will have been demonstrated that investments in education are an important means by which parental education influences child education.

Notably, the predicted values for dichotomous measures fall within the [0, 1] range, and the linear probability model facilitates simple implementation of fixed effects (below). To account for the fact that the error terms are likely correlated across the six different investments, the models are estimated jointly

via Zellner's (1962) seemingly unrelated regression method; because the regressors are identical across investments, this approach is asymptotically equivalent to ordinary least squares.

While omitted variable bias may be mitigated by including the regressors described above, measurement error in educational investments may bias the estimated effect of investments on children's test scores. For example, Robinson (1985) and Hofferth (1999) find that more educated mothers are more likely to overstate time engaged in socially desirable activities such as reading to children. Fortunately, measurement error in the investments included in this study is likely to be minimal. First, nonrequired educational expenditures are constructed from detailed consumption records rather than a single response, a method that has been demonstrated to reduce reporting error (Biemer et al. 2004). Second, whether the household has children's reading material and whether the sampled child has a designated study area were readily observable by survey enumerators; reporting error in these investments should thus be small. Third, parents and children were asked independently about parental reading to children. In most cases, children and parents gave the same answer. In cases where the responses did not match, children were as likely as parents to state that parents read to children, suggesting that any reporting error was not systematic. Similarly, parents and teachers were asked independently whether parents meet with teachers to discuss their children's schooling. As above, the responses matched in the vast majority of cases; where they did not, parents were no more likely than teachers to report that meetings took place, again suggesting that any reporting error is not systematic.⁷ Finally, Juster (1985) finds that reporting error in time allocation is greatly reduced when a given activity is undertaken regularly and with limited variation. If assistance with homework is given frequently and regularly, then reporting error is likely to be small.

IV. Data and Variables

The data come from the Gansu Survey of Children and Families (GSCF),⁸ a survey of 1,970 children ages 9–12 and their families in a province located

⁷ One possible explanation for any difference in responses is that the teacher's questionnaire asked about parent/teacher discussions with parents within the previous month, whereas the mother's questionnaire asked about discussions within the previous year.

⁸ The GSCF was administered in 2000. It was a collaborative effort between researchers in China, Canada, and the United States, including the author.

in northwestern China.⁹ Gansu's 25.3 million people are primarily engaged in agriculture, and the province is broadly reflective of other western provinces in its low income, low educational attainment, and low expenditures on education. The multistage sample drew 20 counties from all nonurban, non-Tibetan counties and 100 villages from those counties.¹⁰ Within each village, children in the correct age range were drawn with equal probability. Separate instruments were administered to children, mothers, heads of household, and village leaders as well as to teachers and principals of children who were enrolled in school at the time of the survey. A cognitive development test was also administered to each child.

Table 1 presents descriptive statistics. The average household spends 46.5 RMB per year on nonrequired educational goods and services for the sampled child.¹¹ Only 6.2% of households allocate less than 10 RMB to educational goods, while 1.8% of households spend at least 200 RMB. Some 54.4% of households have children's books, and 58.8% have designated study areas for children, for example, a child's desk, work table, or bookshelf. On average, mothers and fathers spend 4.1 hours in total helping children with homework each week, although parents do not help their children with homework at all in 32.5% of the sampled households. At the other extreme, 5.7% of households spend at least 14 hours per week helping children with homework.¹² Time spent helping children with homework is inclusive of all children, not just the sampled child. Parents read to the sampled child in almost two-thirds of the surveyed households and discuss the sampled child's school performance with teachers in 76.2% of the surveyed households.

Fathers have completed one grade in junior secondary school on average, while mothers have completed 4.2 primary grades.¹³ Fathers spend all or part

⁹ Of the 1,970 children in the sample, 16 did not reside full time in the sampled village, 17 others were missing important household demographic data such as parental education, and 19 were not enrolled in school at the time of the survey (6 of whom had dropped out of primary school and 1 of whom dropped out after completing primary school). The sample is thus restricted to the 1,918 children who were enrolled in school at the time of the survey, who were full-time residents of the village, and who have complete parent, schooling, and teacher data. For variables common to both the restricted and unrestricted samples, the data are largely indistinguishable.

¹¹ In 2000, US\$1 = 8.27 Chinese RMB.

¹² This statistic is averaged across the entire year and thus properly accounts for parents who migrate for part of the year.

¹⁰ Of Gansu's 86 counties, seven are predominantly Tibetan. These counties were omitted from the sampling because Mandarin is not widely spoken in these areas.

¹³ Although I cannot distinguish between birth parents, adoptive parents, and stepparents, Hofferth and Anderson (2003) find that biology is less important than marital responsibility in fathers' investments in children.

		TABLE 1 SUMMARY STAT				
Variable	Description	Unit	Mean	SD	Min	Max
EXPEND	Nonrequired					
	expenditures	RMB	46.519	55.595	0	836
BOOKS	Has children's books	Dummy	.544	.498	0	1
DESK	Has designated					
	study area	Dummy	.588	.492	0	1
HELP	Time helping with					
	homework	Hours/week	4.121	4.953	0	35
READ	Parents read to					
	child	Dummy	.657	.475	0	1
DISCUSS	Discuss schooling					
	with teacher	Dummy	.762	.426	0	1
FED	Father's education	Grades	6.985	3.515	0	15
MED	Mother's education	Grades	4.190	3.514	0	12
FAGE	Father's age	Years	37.411	4.846	27	57
MAGE	Mother's age	Years	35.060	4.210	25	55
FRES	Father's village					
	residency	Months/year	9.935	3.475	0	12
MRES	Mother's village					
	residency	Months/year	11.732	1.547	0	12
MALE	Male child	Dummy	.539	.499	0	1
AGE	Child's age	Years	11.019	1.069	9	12.9
GRADE	Child's grade	Current level	4.301	1.343	1	9
ENROLLED	Other enrolled					
	children	Number	.866	.714	0	4
NONENR	Nonenrolled					
	children	Number	.452	.638	0	4
WEALTH	Household wealth	RMB	14,773.81	16,963.81	115	209,740
TCHRRANK	Teacher quality					
	ranking	0 = probation,				
		1 = rank 1,				
		2 = rank 2,			_	
		3 = highest	1.468	.953	0	3
HICOG	Above average cog-	_			_	
	nitive score	Dummy	.490	.500	0	1

of 2 months working outside the village on average, although the median father does not migrate at all. Fewer than 4% of women migrate for work, and both parents are absent for the entire year in only six households. The present value of housing and other durables (a measure indicative of household wealth) averages 14,773.8 RMB, but there is considerable variation. For example, 3% of households possess over 50,000 RMB in durables.

Boys constitute 53.9% of the sample. Primary school enrollees account for 96.0% of the sample, an artifact not only of the ages of the sampled children but also of the delayed age of enrollment prevalent in Gansu. The median child is in fourth grade, having enrolled at age 7. The household has 0.9 other enrolled children and 0.4 nonenrolled children, on average.

Teachers are regularly and systematically evaluated on such disparate mea-

		Parental Education Quartiles				
Investment	Unit	1	2	3	4	
EXPEND	RMB	33.97	47.37	52.10	54.01	
BOOKS	Dummy	.38	.52	.64	.64	
DESK	Dummy	.47	.54	.64	.73	
HELP	Hours/week	2.24	3.91	4.90	5.72	
READ	Dummy	.50	.63	.71	.82	
DISCUSS	Dummy	.65	.75	.79	.88	

 TABLE 2

 PARENTAL EDUCATION AND EDUCATION-RELATED INVESTMENTS

sures as education, experience, attendance, professional publications, student test performance, and student and peer evaluations (Park and Hannum 2001). Because this official quality ranking incorporates many aspects of teacher quality, it is likely to be more informative than measures of teacher quality used in other countries. Moreover, teachers in the surveyed areas follow student cohorts through school, so controlling for current teacher quality goes a long way toward controlling for the quality of teachers in former grades.

The child cognitive development test was developed by researchers at the Institute for Psychology at the Chinese Academy of Social Sciences and was designed to be independent of academic achievement. This measure is employed to control both for innate ability and expected returns to child education. Scores on the cognitive development test do vary significantly by age, however, so test scores were converted into Z-scores by age measured in half-year increments.

Finally, it should be noted that China's system of restrictive residency laws prevents rural residents from legally attending school outside their registered residence areas. Since most villages have only one school, village fixed effects control for school quality as well as local socioeconomic conditions, labor market opportunities, and other forms of unobserved heterogeneity.

V. Investment in Goods and Services

Table 2 presents cross-tabulations of parental education and educational investments. Neglecting other factors, investments in both goods and time rise monotonically with parental education, suggesting that more educated parents make greater investments in both goods and time.

A. Estimates

The effect of father's and mother's education on the provision of six different education-related investments is estimated via seemingly unrelated regression methods. Note that because the regressors are the same across all six specifications, seemingly unrelated regression is asymptotically equivalent to

	DETER	-	ABLE 3 EDUCATIONAL I	NVESTMENTS		
	EXPEND	BOOKS	DESK	HELP	READ	DISCUSS
Estimates:						
FED	.0228***	.0152***	.0103**	.1986***	.0093**	.0056
	(2.59)	(3.09)	(2.10)	(4.08)	(2.00)	(1.31)
MED	.0332***	.0173***	.0143***	.2706***	.0261***	.0173***
	(3.62)	(3.39)	(2.81)	(5.33)	(5.37)	(3.88)
FAGE	.0088	0023	0003	0296	0090**	.0044
	(1.33)	(.63)	(.09)	(.81)	(2.58)	(1.35)
MAGE	0035	.0129***	.0021	0474	.0085**	0028
	(.45)	(3.03)	(.50)	(1.12)	(2.10)	(.75)
MALE	.0452	.0200	.0420	.0867	0033	0308
	(.50)	(.040)	(.48)	(.17)	(.07)	(.70)
AGE	.0148	0442 ^c	0454***	2144	0184	0206*
	(.60)	(3.25)	(3.34)	(1.59)	(1.42)	(1.73)
ENROLLED	.4460***	0657***	0062	162	0358**	.0044
	(14.10)	(3.74)	(.35)	(.93)	(2.14)	(.28)
NONENR	.009	0764***	027	0460	0302	.0180
	(.25)	(3.78)	(1.34)	(.23)	(1.57)	(1.02)
FED × MALE	.0044	0013	0036	0554	.0029	.0076
	(.36)	(.20)	(.54)	(.83)	(.46)	(1.30)
MED × MALE	0147	0052	.0046	.0421	.0025	.0002
	(1.21)	(.77)	(.68)	(.63)	(.39)	(.03)
Constant	1.3456***	.3397	1.1130***	8.1775***	.6245***	.6983***
	(3.62)	(1.64)	(5.38)	(3.98)	(3.17)	(3.86)
Hypotheses:						
1. H0: FED = 0						
for boys	10.28***	8.61**	2.00	9.34***	7.47***	10.34***
2. H0: MED =						
0 for boys	4.94*	6.85**	16.66***	46.08***	42.00***	18.62***
3. H0: FED =						
MED for girls	.50	.07	.25	.79	4.69**	2.71
4. H0: FED =						
MED for						
boys	.39	.05	2.56	4.96**	5.02**	.40

Note. Grade dummies are included; for estimates, *t*-statistics are in parentheses; for hypotheses, *F*-statistics are shown.

* Significant at the 10% level.

** Significant at the 5% level.

*** Significant at the 1% level.

ordinary least squares regression. Estimates from the parsimonious model (eq. [1]), which controls for parental age, child gender and age, and other children in the household, is presented in table 3.

Parental education has a strongly significant effect on the provision of education-related goods for daughters. An additional year of father's education (FED) increases predicted spending on nonrequired educational goods (EX-PEND) for daughters by 2.3% (significant at the 0.01 level). Similarly, an additional year of father's education raises the probability that the household has children's reading materials (BOOKS) by 1.5 percentage points (2.8% of

Brown 771

the mean, significant at the 0.01 level) and the probability that a daughter has a designated study area (DESK) by 1.0 percentage points (1.8% of the mean, significant at the 0.05 level). The effect of mother's education (MED) on the provision of education-related goods for daughters is somewhat larger: an additional year of completed education raises predicted education-related spending by 3.3%, increases the probability that the household has children's books by 1.7 percentage points, and increases the probability that the daughter has a designated study area by 1.4 percentage points (all significant at the 0.01 level). Although all of the point estimates are larger for mothers than for fathers, *F*-tests show that the hypothesis that there are effects of father's and mother's education on goods investments for daughters cannot be rejected.

Interactions of parental education and a male child dummy (MALE) are not statistically significant, suggesting the absence of systematic gender bias in goods investments on the part of both mothers and fathers. Specifically, an additional year of father's education increases predicted expenditures by 2.7% (significant at the 0.01 level) for sons and increases the probability that children's books are available in the household by 1.4 percentage points (significant at the 0.01 level). However, an additional year of father's education does not have a statistically significant effect on the likelihood that the son has a designated study area. The effect of mother's education on education-related expenditures and on the provision of books is smaller for sons than for daughters, although the difference is not statistically significant. An additional year of mother's education raises the probability of books being provided by 1.2 percentage points (significant at the 0.01 level). By contrast, the effect of an additional year of mother's education on the probability that a child has a designated study area is larger for sons (1.9 percentage points, significant at the 0.01 level) than for daughters, although the difference is again not statistically significant. As with daughters, I cannot reject the hypothesis that father's and mother's education have statistically identical effects on goods investments in sons' education.

Older students are less likely to have access to certain educational goods: an additional year of age reduces the probability that a household has children's books by 4.4% and reduces the probability that a child has a designated study area by 4.5% (both significant at the 0.01 level). This suggests that poorer parents may delay their children's enrollment or that parents with less favorable views of education may delay enrollment and may provide fewer investments. Alternatively, older students may perform better in school, and hence the return to such investments may be low. The presence of other children in the household (ENROLLED and NONENR)—whether or not they are enrolled in school—similarly reduces the likelihood that a household has children's

reading material by 6.6%–7.6% (significant at the 0.01 level), suggesting that additional children crowd out investment in educational goods. Interestingly, there is no statistically significant effect of additional children on the probability that a child has a designated study area. An additional child enrolled in school increases spending on nonrequired educational goods by 44.6% (significant at the 0.01 level), while nonenrolled children have no such effect, suggesting either that some of the nonrequired expenditures are strongly encouraged or that children's reading materials are not seen as being as valuable an investment as other education-related goods.

Similar results are found for parental investments in time, yet some important differences are evident. An additional year of father's education increases the time allocated to helping children with homework each week (HELP) by 0.20 hours when the sampled child is a daughter, 4.8% of the mean (significant at the 0.01 level). An additional year of mother's education raises the amount of time allocated to helping children with homework by 0.27 hours when the sampled child is a daughter (significant at the 0.01 level). However, we cannot reject the hypothesis that the marginal effect of education on time allocated to helping children with homework is statistically identical for mothers and fathers. More educated parents are more likely to read to their daughters (READ), with an additional year of father's education increasing the probability by 0.9 percentage points, 1.4% of the mean (significant at the 0.05 level) and an additional year of mother's education increasing the probability by 2.6 percentage points (significant at the 0.01 level). Moreover, results of an F-test show that the effect of parental education on the level of investment differs between mothers and fathers (significant at the 0.05 level), suggesting that mother's education leads to more time investments in children's education. An additional year of father's education does not significantly affect the probability that parents discuss a daughter's schooling with her teachers (DISCUSS), but an additional year of mother's education raises this probability by 1.7 percentage points, 2.3% of the mean (significant at the 0.01 level). The difference in the estimated effect of parental education is statistically significant at the 0.10 level, suggesting again that mother's education leads to disproportionate time investments in children's education.

As with goods investments, the interaction of parental education and the male dummy does not yield statistically significant estimates, suggesting that gender bias in time investments is very minor. An additional year of father's education raises the weekly time that parents allocate to helping children with homework by 0.14 hours when the sampled child is male, raises the probability that parents read to a son by 1.2 percentage points, and raises the probability that parents discuss a son's schooling with his teachers by 1.3% (all significant

at the 0.01 level). An additional year of mother's education raises the weekly time that parents allocate to helping children with homework by 0.31 hours when the sampled child is male, raises the probability that parents read to a son by 2.9 percentage points, and raises the probability that parents discuss a son's schooling with his teachers by 1.8 percentage points (all significant at the 0.01 level). The effect of mother's education on the time allocated to helping children with homework and the probability the parents read to sons is larger than that of father's education (both significant at the 0.05 level), again suggesting that mother's education has a larger effect on time investments in education than father's education.

Child age has a negative effect on the probability that parents discuss their children's schooling with teachers, with an additional year reducing the probability by 2.1 percentage points (significant at the 0.10 level). As discussed above, it may be the case that poor parents can afford neither to enroll their children on time nor to leave work to talk with children's teachers. Alternatively, this may reflect lower interest on the part of parents or older children doing sufficiently better in school that this investment is not warranted. The number of other children reduces the probability that parents read to the sampled child (significant at the 0.05 level for other children enrolled in school), again suggesting that other children may vie for parent's time. Curiously, the number of children does not have a statistically significant effect on the time that parents allocate to helping with homework; the number of older siblings is negatively correlated with the time that parents devote to helping with homework, suggesting that older siblings substitute for parents in helping younger children.

It is thus evident that more parental education is associated with larger educational investments in both goods and time. Moreover, while higher mother's and father's education lead to similar increases in goods investments, additional education on the part of the mother leads to disproportionately higher time investments. Finally, there is no statistical evidence suggesting that either parent favors children of one gender over the other when it comes to making education-related investments.

B. Robustness

While the parsimonious model highlights the relationship between parental education and investments in children's learning across a variety of investments, it does not account for other important determinants of educational investments. As discussed in Section III, more educated parents may have fewer resource constraints; thus controlling for household wealth may isolate the true relationship between parental education and investments in children's

schooling. Similarly, more educated parents invest differently in their children's education to compensate for teacher and school quality; if more educated parents are better able to assess teacher competence or school quality, then controlling for these attributes may better identify the relationship between parental education and educational investments. Finally, a child's cognitive development may affect the optimal investment because more educated parents may invest more in very gifted children. Alternatively, more educated parents may help less gifted children by providing greater investments in their schooling. These three possibilities are explored in tables 4–7.

The logged value of household durables (WEALTH) has a strong and positive effect on educational expenditures (table 4). The elasticity of wealth with respect to nonrequired educational expenditures is 0.11 (significant at the 0.01 level). Similarly, a 1% increase in the value of durables increases the probability that the household has children's reading materials by 0.11 percentage points and increases the probability that the household has a designated study area for the child by 0.09 percentage points (both significant at the 0.01 level). The value of household durables does not significantly affect investments in time.

Controlling for the value of household durables has little effect on the point estimates for parental education. For example, including the wealth measure reduces the estimated effect of an additional year of father's education on expenditures for daughters from 2.3% to 1.9%, remaining significant at the 0.05 level. Indeed, for every investment except time allocated to helping children with homework (owing to a weakly negative relationship between wealth and time spent helping with homework), the point estimates fall slightly when controlling for wealth. With the exception of the effect of father's education on the probability that the household has a designated study area for daughters, these effects also remain statistically significant. An additional year of mother's education raises the probability that a household has a designated study area for a sampled son by 1.6 percentage points, whereas an additional year of father's education raises this probability by only 0.3 percentage points, a difference that is significant at the 0.10 level. Otherwise, the differential effects of mother's and father's education remain as above. Note again that there is no statistical evidence in favor of gender bias in any investment.

With the inclusion of village fixed effects to account for differences in school quality (as well as unobserved heterogeneity at the local level), the teacher quality variable (TCHRRANK) does not have a statistically significant effect on the provision on educational investments (table 5). This suggests that poor teacher quality is not the primary concern of parents who make educational

Brown 775

DETER	RMINANTS OF EI		ABLE 4 NVESTMENTS V		FOR WEALTH	
	EXPEND	BOOKS	DESK	HELP	READ	DISCUSS
Estimates:						
FED	.0193**	.0115**	.0073	.2033***	.0090*	.0053
	(2.19)	(2.39)	(1.51)	(4.16)	(1.92)	(1.22)
MED	.0291***	.0131***	.0109**	.2760***	.0257***	.0169***
	(3.18)	(2.61)	(2.16)	(5.42)	(5.27)	(3.77)
FAGE	.0125*	.0015	.0028	0345	0087**	.0047
	(1.88)	(.41)	(.75)	(.93)	(2.47)	(1.46)
MAGE	0066	.0097**	0005	0433	.0083**	0031
	(.86)	(2.32)	(.11)	(1.02)	(2.03)	(.83)
MALE	.045	.0197	.0419	.0870	0033	0308
	(.50)	(.40)	(.85)	(.18)	(.07)	(.70)
AGE	.0220	0366***	0393***	2241*	0178	0198*
	(.91)	(2.74)	(2.92)	(1.66)	(1.37)	(1.67)
ENROLLED	.4515***	0601***	0016	1692	0353**	.0049
	(14.35)	(3.48)	(.09)	(.97)	(2.11)	(.32)
NONENR	.0152	0700***	0218	0543	0297	.0186
	(.42)	(3.53)	(1.09)	(.27)	(1.54)	(1.05)
FED × MALE	.0038	0019	0041	0547	.0029	.0076
	(.32)	(.29)	(.62)	(.82)	(.45)	(1.29)
MED × MALE	0136	004	.0055	.0407	.0026	.0003
	(1.13)	(.61)	(.83)	(.61)	(.40)	(.05)
WEALTH	.1055***	.1092***	.0889***	1408	.0094	.0111
	(4.71)	(8.90)	(7.19)	(1.13)	(.79)	(1.01)
Constant	.3778	6621***	.2977	9.4688***	.5384**	.5968***
	(.89)	(2.85)	(1.27)	(4.03)	(2.39)	(2.89)
Hypotheses:						
1. H0: FED = 0						
for boys	7.43***	4.28**	.48	9.96***	6.96***	9.58***
2. H0: MED =						
0 for boys	3.52*	3.98**	12.85**	46.98***	40.99***	17.87***
3. H0: FED =						
MED for girls	.46	.04	.20	.81	4.67**	2.69
4. H0: FED =						
MED for						
boys	.30	.00	3.05*	4.87**	5.08**	.42

Note. Grade dummies are included; for estimates, *t*-statistics are in parentheses; for hypotheses, *F*-statistics are shown.

* Significant at the 10% level.

** Significant at the 5% level.

*** Significant at the 1% level.

investments.¹⁴ The addition of these controls reduces the point estimates for father's education on all six educational investments for both sons and daughters. For example, an additional year of father's education raises nonrequired expenditure for a son by 2.0% when controlling for teacher quality and village

¹⁴ An alternative explanation that teacher quality varies little within individual schools in Gansu has been ruled out by Park and Hannum (2001). Teacher rank does have a modest effect on investments in the absence of village fixed effects, but the other estimates are not appreciably different from those in the parsimonious model described above.

	ANTS OF EDUCA		GE FIXED EFFE			
	EXPEND	BOOKS	DESK	HELP	READ	DISCUSS
Estimate:						
FED	.0197**	.0078*	.0094**	.1817***	.0067	0006
	(2.46)	(1.69)	(2.06)	(3.79)	(1.48)	(.15)
MED	.0097	.0091	.0075	.2258***	.0189***	.0108**
	(1.12)	(1.84)	(1.52)	(4.39)	(3.88)	(2.41)
FAGE	.0086	0018	0003	0699*	0098***	.0027
	(1.41)	(.52)	(.08)	(1.92)	(2.85)	(.85)
MAGE	.0036	.0062	.0009	0313	.0046	.0003
	(.51)	(1.54)	(.22)	(.74)	(1.14)	(.09)
MALE	.0672	.0349	.0435	.0654	0023	0577
	(.84)	(.76)	(.95)	(.14)	(.05)	(1.37)
AGE	0035	0214	0377***	.0056	0057	0143
	(.15)	(1.59)	(2.82)	(.04)	(.43)	(1.17)
ENROLLED	.4772***	.0124	.0215	0401	.0191	.0057
	(15.47)	(.70)	(1.22)	(.22)	(1.09)	(.35)
NONENR	.0250	0117	0192	0910	.0052	.0001
	(.74)	(.60)	(.99)	(.45)	(.27)	(.01)
FED × MALE	.0001	.0029	0029	0277	.0035	.0106*
	(.01)	(.47)	(.46)	(.43)	(.58)	(1.88)
MED × MALE	0063	0102	.0014	0106	.003	0006
	(.57)	(1.62)	(.22)	(.16)	(.48)	(.10)
TCHRRANK	0025	.0124	.0099	0124	.0110	.0036
	(.11)	(.93)	(.75)	(.09)	(.83)	(.30)
Constant	1.6364***	.7070***	1.2296***	5.6610**	.7545***	.8029***
	(4.13)	(3.11)	(5.45)	(2.39)	(3.36)	(3.88)
Hypotheses:	(()	()	(()	()
1. H0: FED = 0						
for boys	6.56**	5.76**	2.22	11.07***	5.48**	6.17**
2. H0: MED =	0.00	5.70	2.22	11.07	0.10	0.17
0 for boys	.18	.06	3.89**	20.83***	24.07**	6.23**
3. H0: FED =		.00	0.07	20.00	21.07	0.20
MED for girls	.57	.03	.07	.30	2.59	2.69
4. H0: FED = MED for	.57	.00	.07	.50	2.37	2.07
boys	1.69	2.61	.10	.65	2.63	.00
buys	1.07	2.01	.10	.05	2.05	.00

TABLE 5
DETERMINANTS OF EDUCATIONAL INVESTMENTS WITH CONTROLS FOR TEACHER QUALITY

Note. Grade dummies are included; for estimates, *t*-statistics are in parentheses; for hypotheses, *F*-statistics are shown;

* Significant at the 10% level.

** Significant at the 5% level.

*** Significant at the 1% level.

characteristics versus 2.7% in the parsimonious specification. Nevertheless, the effect remains positive and significant (generally at the 0.05 level or above) for the same investments for both sons and daughters, with the exception of reading to daughters.

The effects of mother's education on the probability that the household owns children's reading materials and on the probability that sons have designated study areas remain positive and statistically different from zero (albeit

Brown 777

with reduced point estimates). By contrast, the effect of mother's education on goods investments virtually disappears when controlling for teacher quality and school and village characteristics. For example, an additional year of mother's education increases expenditures for boys by 1.9% in the parsimonious specification but increases expenditures for boys by only 0.3% with the additional controls. When excluding father's education from this fixed effects regression, however, the effect of mother's education on goods investment is strong, positive, and significant, suggesting that the reductions in statistical significance are due to multicollinearity between mother's and father's education. However, the effect of mother's education on investments in time remains positive and highly significant (at the 0.05 level or above) even in the original specification in which father's education is included as a regressor.

The hypothesis that the effects of mother's and father's education on educational investments are equal cannot be rejected in this specification. Nevertheless, the hypotheses stating that mother's and father's education have equal effects on the probabilities that sons have access to reading material, that parents read to sons, that parents read to daughters, and that parents discuss schooling with daughters' teachers are nearly significant with p-values ranging between 0.101 and 0.108. Thus, the pattern described above in which mother's education results in disproportionate investments in time for children of both sexes persists. Moreover, it appears that father's education results in somewhat greater goods investments for sons, but not daughters.

Finally, there is some weak evidence for preferential treatment of sons by fathers in this specification. An additional year of father's education does not have any discernible effect on the probability that parents discuss a daughter's education with her teachers, but it does increase the probability that parents discuss a son's schooling with his teachers by 1.0%, a difference that is significant at the 0.10 level. This provides weak statistical evidence for systematic gender bias in educational investments.

High cognitive development among children (HICOG) has a weakly positive direct effect on educational investments undertaken by parents (table 6). Children with above-average cognitive development are 19.8 percentage points more likely to have access to children's reading material (significant at the 0.01 level) and are 17.5 percentage points more likely to have parents who read to them (significant at the 0.10 level), suggesting that parents reward children with higher cognitive development. Cognitive development does not have a statistically significant direct effect on other investments.¹⁵

More educated parents make greater educational investments even when

¹⁵ None of the foregoing results is sensitive to different cut-offs for high cognitive development.

DE	DETERMINANTS OF EDUCATIONAL INVESTMENTS WITH CONTROLS FOR CHILD COGNITIVE DEVELOPMENT	IONAL INVESTMENTS WI	S WITH CONTROLS FOR CHIL	D COGNITIVE DEVELOP	ИЕИТ	
	EXPEND	BOOKS	DESK	HELP	READ	DISCUSS
Estimates:						
FED	.0197	.0201***	.0127**	.1346***	.0176**	.0078
	(1.70)	(3.13)	(1.97)	(2.09)	(2.86)	(1.39)
MED	.0380***	.0160**	.0118*	.2873***	.0236***	.0083
	(2.99)	(2.26)	(1.66)	(4.08)	(3.51)	(1.35)
FAGE	.0088	0022	0003	0301	0090***	.0042
	(1.33)	(.59)	(.09)	(.82)	(2.57)	(1.30)
MAGE	0016	.0115***	.0019	0524	.0081*	0020
	(.21)	(2.69)	(.44)	(1.23)	(2.00)	(.54)
MALE	.1301	.0289	.0387	5410	.0673	0386
	(1.11)	(.44)	(.59)	(.83)	(1.08)	(.67)
AGE	.0058	0345**	0447***	1801	0141	0258**
	(.23)	(2.51)	(3.24)	(1.31)	(1.08)	(2.14)
ENROLLED	.4417***	0631***	0052	1476	0361**	.0023
	(13.94)	(3.59)	(.29)	(.84)	(2.16)	(.15)
NONENR	.0054	0748***	0266	0245	0314	.0162
	(.15)	(3.71)	(1.31)	(.12)	(1.63)	(.92)
FED × MALE	6000.	0049	0025	.0031	0105	.0038
	(.05)	(.54)	(.27)	(.03)	(1.22)	(.47)
MED × MALE	0317*	.0058	.0095	.0831	.0127	0600.
	(1.84)	(.61)	(.99)	(.87)	(1.39)	(1.08)
HICOG	1066	.1979***	.0545	6712	.1750**	858
	(.80)	(2.67)	(.73)	(16.)	(2.48)	(1.32)
HICOG × FED	.0070	0115	0060	.1504	0192**	0055
	(.40)	(1.18)	(.61)	(1.55)	(2.06)	(.64)
HICOG × MED	0101	.0018	.0047	0374	.0047	.0188**
	(.56)	(.18)	(.46)	(.37)	(.49)	(2.14)
HICOG × MALE	1978	0294	0017	1.5174	1714*	.0170
	(1.08)	(.29)	(.02)	(1.50)	(1.77)	(.19)

TABLE 6 ONAL INVESTMENTS WITH CONTROLS FOR CHILD COGNIT

HICOG × FED × MALE	.0080	.0085	0014	1433	.0303**	.0089
	(.33)	(.63)	(.10)	(1.06)	(2.35)	(.75)
HICOG × MED × MALE	.0342	0204	0083	0791	0196	0182
	(1.40)	(1.51)	(.61)	(.59)	(1.52)	(1.53)
Constant	1.3790***	.2460	1.1022***	8.3091***	.5392***	.7440***
	(3.66)	(1.18)	(5.25)	(3.99)	(2.71)	(4.06)
Hypotheses:						
1. H0: FED = 0 for boys	3.14*	5.59**	2.52	4.60**	1.33	4.24**
2. H0: MED = 0 for boys	.28	10.60***	10.01***	30.76***	32.47***	8.78***
3. H0: FED = MED for girls	.87	.15	.01	1.96	.34	00.
4. H0: FED = MED for boys	2.31*	12.20***	9.08***	24.72***	21.75***	9.78***
5. H0: FED = 0 for HICOG						
girls	4.03**	1.37	.83	14.98***	.05	.14
6. H0: MED = 0 for HICOG						
girls	4.59**	6.02**	5.12**	11.97***	16.87***	18.24***
7. H0: FED = 0 for HICOG						
boys	.83	.03	00.	2.84*	2.64	.14
8. H0: MED = 0 for HICOG						
boys	.07	69.	1.82	1.01	.10	60.
9. H0: FED = MED for HI-						
COG girls	00.	.57	.64	60.	6.77***	5.47**
10. H0: FED = MED for HI-						
COG boys	.07	69.	1.82	1.01	.10	60.
Note. Grade dummies are included; for estimates, t-statistics are in parentheses; for hypotheses, F-statistics are shown. * Significant at the 10% level.	r estimates, t-statistics	are in parentheses; f	or hypotheses, F-statist	ics are shown.		

> Significant at the 5% level. ** Significant at the 5% level. *** Significant at the 1% level.

controlling for child cognitive development. For example, an additional year of father's education raises the probability that parents provide reading materials to children who do not score above average on the cognitive development test by between 1.5 percentage points (significant at the 0.05 level) and 2.0 percentage points (significant at the 0.01 level). The effect of father's education is not significant for either sons or daughters with high cognitive development. An additional year of mother's education increases the probability that parents provide reading materials to daughters who do not score highly by 1.6 percentage points and to sons who do not score highly by 2.2 percentage points (both significant at the 0.01 level). Furthermore, mother's education increases the probability that high-scoring daughters have access to reading materials in the home by 1.8 percentage points (significant at the 0.05 level). The effect of mother's education on providing reading material to high-scoring sons is negligible.

Results for time investments are similar. An additional year of father's education raises the probability that parents read to the sampled child by 1.8 percentage points for daughters who do not score highly on the cognitive development test (significant at the 0.05 level); the effect of father's education for low-scoring sons and for high-scoring sons and daughters is not significantly different from 0. An additional year of mother's education raises the probability that parents read to the sampled child by 2.4 percentage points for daughters who do not score highly, by 3.6 percentage points for sons who do not score highly, and by 2.8 percentage points for high-scoring daughters (all significant at the 0.01 level). Mother's education has no effect for high-scoring sons. Indeed, mother's education is associated with significantly higher investments for all gender-ability combinations with the exception of high-scoring boys; father's education has a significant effect on four of the six investments for low-scoring boys and girls and on two of the six investments for high-scoring girls.

With the inclusion of the cognitive development measure, the statistical differences in the effects of father's and mother's education are more pronounced. For example, an additional year of father's education raises predicted expenditures on educational goods by 2.1% for sons who have not scored above average on the cognitive development test, while an additional year of mother's education raises predicted spending by 0.6%, a difference that is statistically significant at the 0.10 level. However, the effect of mother's education on investments for sons who have not scored highly on the cognitive test is significantly larger than that of father's education for each of the other five investments. For example, an additional year of father's education raises the time allocated to helping a son with homework by 0.14 hours while an

additional year of mother's education raises the time allocated by 0.37 hours, a difference that is significant at the 0.01 level. Interestingly, there is no discernible difference in the effect of father's and mother's education for daughters who have not scored highly on the cognitive test or for sons who have. For high-scoring daughters, however, mother's education has a larger effect than father's education for both the probability that a parent reads to the sampled daughter and the probability that parents discuss schooling with her teachers. Overall, these results suggest that the effect of mother's education on investment in both goods and time is larger than that of father's education and that this difference is especially pronounced for time investments.

It is also interesting to note that the interaction term of mother's education and the male child dummy is negative in education-related expenditures (significant at the 0.10 level), suggesting that more educated mothers favor lowscoring daughters over low-scoring sons. By contrast, there is evidence to suggest that more educated fathers favor high-scoring sons over high-scoring daughters in terms of reading to children (significant at the 0.05 level). These findings again suggest that some mild gender bias may exist in educational investments.

Table 7 presents results from regressions in which wealth, teacher quality, child cognitive development, and village fixed effects are all added to the parsimonious specification. Interestingly, any statistical significance of the high cognitive development measure disappears when fixed effects are included, suggesting that the cognitive development score correlates with some unobserved characteristic of the village.¹⁶ Nevertheless, the inclusion of village fixed effects reduces the estimated effect of mother's education on investment in goods, likely as a result of multicollinearity with father's education.¹⁷ Still, mother's education has a small negative effect (significant at the 0.05 level) on the probability that a son who has not scored highly on the cognitive development test has a study area. The negative coefficient suggests that mother's education correlates with some unobserved characteristic of the village that affects certain expenditures.¹⁸ Parental education remains a positive and significant determinant of most of the time investments for children who scored below average on the cognitive development test. However, father's

¹⁶ For example, some communities may have norms about children's diets, which have been shown to affect cognitive development in numerous developing countries (e.g., Freeman et al. 1980).

¹⁷ The point estimates for mother's education are largely positive and significant when father's education is omitted from the regression.

¹⁸ If more educated mothers are more likely to live in villages in which schools are open late to promote studying, then more educated mothers may substitute studying at school for studying at home.

	DETERMINANTS (DF EDUCATIONAL INVES	DETERMINANTS OF EDUCATIONAL INVESTMENTS WITH A FULL SET OF CONTROLS	ET OF CONTROLS		
	EXPEND	BOOKS	DESK	НЕГР	READ	DISCUSS
Estimates:						
FED	.0196*	.0058	.0104*	.1331**	.0151**	.0051
	(1.86)	(96)	(1.74)	(2.11)	(2.53)	(.92)
MED	.0141	.0078	.0008	.2256***	.0166**	.0085
	(1.18)	(1.15)	(.11)	(3.16)	(2.46)	(1.36)
FAGE	.0104*	0000	.0014	0685*	0099***	.0027
	(1.70)	(.01)	(.40)	(1.87)	(2.85)	(.84)
MAGE	.0025	.0048	0003	0336	.0047	.0003
	(.35)	(1.19)	(.08)	(.79)	(1.17)	(.07)
MALE	.1280	.0312	.0316	2468	.0669	0431
	(1.21)	(.22)	(.53)	(.39)	(1.12)	(.78)
AGE	0034	0178	0375***	.0450	0038	0146
	(.14)	(1.31)	(2.78)	(.32)	(.28)	(1.17)
ENROLLED	.4714***	.0062	.0160	0513	.0171	.0050
	(15.22)	(.35)	(.91)	(.28)	(.97)	(.31)
NONENR	.0212	0157	0224	0996	.0024	0008
	(.62)	(.81)	(1.16)	(.49)	(.13)	(.04)
FED × MALE	0044	0011	0063	0153	0118	.0045
	(.30)	(.13)	(.76)	(.17)	(1.43)	(.59)
MED × MALE	0199	.0016	0121	.0539	.0130	.0065
	(1.28)	(.18)	(1.37)	(.58)	(1.47)	(.80)
WEALTH	.0594***	.0675***	.0612***	.0602	.0022	.0046
	(2.61)	(5.20)	(4.74)	(.44)	(.17)	(.38)
TCHRRANK	0047	.0124	.0088	0035	.0133	.0043
	(.20)	(.94)	(.67)	(.03)	(1.01)	(.36)
HICOG	0536	.0365	.0147	2906	.1328	.0685
	(.42)	(.51)	(.20)	(.38)	(1.85)	(1.04)
HICOG × FED	0034	0002	0069	.1607	0197**	0136
	(.21)	(.02)	(.76)	(1.12)	(2.18)	(1.63)

TABLE 7 ITS OF EDUCATIONAL INVESTMENTS WITH A FULL SET OF

HICOG × MED	0129	0012	6600.	0095	.0056	.0058
	(.78)	(.12)	(1.05)	(.10)	(.59)	(.67)
HICOG × MALE	1383	.0413	.0166	.7086	1650*	.0369
	(.84)	(.44)	(.18)	(.72)	(1.77)	(.43)
HICOG × FED × MALE	0082	.0069	.0072	0395	.0343**	.0140
	(.37)	(.55)	(.58)	(.30)	(2.77)	(1.23)
HICOG × MED × MALE	.0271	0213*	0198	1163	0206	0147
	(1.22)	(1.69)	(1.58)	(.88)	(1.64)	(1.28)
Constant	1.0585**	.0153	.6303**	4.7949*	.6388**	.7248***
	(2.30)	(90)	(2.41)	(1.74)	(2.45)	(3.01)
Hypotheses:						
1. H0: FED $=$ 0 for boys	2.03	09.	.46	3.38*	.29	2.93*
2. H0: MED $= 0$ for boys	.26	2.10	3.97**	16.85***	21.12***	6.32**
3. H0: FED = MED for girls	.10	.04	.90	.74	.02	.13
4. H0: FED = MED for boys	1.02	1.85	2.87*	13.59***	12.57***	6.49***
5. H0: FED = 0 for HICOG						
girls	1.80	.65	.26	10.89***	.44	1.80
6. H0: MED = 0 for HICOG						
girls	.01	.93	2.39	8.83***	10.42***	5.05**
7. H0: FED = 0 for HICOG						
boys	.54	.33	.05	.86	2.62	.10
8. H0: MED = 0 for HICOG						
boys	.63	60.	.04	<i>TT.</i>	2.63	.13
9. H0: FED = MED for HI-						
COG girls	.58	.01	.40	.04	5.64**	4.79**
10. H0: FED = MED for HI-						
COG boys	.43	5.83**	.02	.09	.11	.19
Note. Grade dummies are included;	for estimates, t-statistic	cs are in parentheses;	included; for estimates, t-statistics are in parentheses; for hypotheses, F-statistics are shown	stics are shown.		

* Significant at the 10% level. ** Significant at the 5% level. *** Significant at the 1% level.

783

TABLE 8
PERCENTAGE OF MOTHERS WHO AGREE THAT EDUCATION HAS "A GREAT DEAL OF
INFLUENCE" ON A CHILD'S FUTURE INCOME

	School v	econdary ersus Pri- School	Senior Secondary School versus Junior Secondary School	
	Boys	Girls	Boys	Girls
Lowest quintile of MED Highest quintile of MED	41.4 48.7	40.6 47.1	51.1 58.2	46.1 51.8

education does not have a statistically significant impact on the probability that parents read to high-scoring daughters or that parents discuss a highscoring daughter's schooling with her teachers, and parental education does not have a statistically significant effect on time investments in high-scoring sons.

With all of the regressors included, father's education is found to disproportionately affect the probability that high-scoring sons have access to reading materials in the home. By contrast, the estimated effect of mother's education is larger than that of father's education for time investments in sons who did not score highly on the cognitive development test and for daughters who did. Mother's education also disproportionately affects the probability that sons who did not score highly have designated study areas. Again, these findings suggest that the effect of mother's education is at least as important as father's education in determining educational investment, especially with regards to investments in time. Finally, there is again some limited evidence of gender bias; more educated fathers favor high-scoring sons over high-scoring daughters when reading to children, more educated mothers favor high-scoring daughters over high-scoring sons when providing reading materials to children.

C. Discussion

As described in the introduction, the finding that both types of educational investment rise with parental education suggests that more educated parents perceive that the returns to education are higher for their children, that more educated parents have different preferences for children's education, or both. Distinguishing between these scenarios is difficult empirically, although there is some evidence in favor of both being true.

First, mothers were asked about the expected income difference if their children obtained a junior secondary education versus a primary education, and if they obtained a senior secondary education versus a junior secondary education (table 8). Among mothers in the lowest quintile of education, 41.4%

believe that junior secondary education has "a great deal" of influence for the future salary of boys, and 40.6% of mothers believe that junior secondary education has "a great deal" of influence for the future salary of girls. Among mothers in the highest quintile of education, 48.7% believe that junior secondary education has "a great deal" of influence for the future salary of boys, and 47.1% of mothers believe that junior secondary education has "a great deal" of influence on the future salary of girls. Some 51.1% of low education mothers thought that senior secondary schooling makes a large difference for boys, while 58.2% of high education mothers did. The corresponding figures for girls at the senior secondary level are 46.1% and 51.8%. Thus, more educated mothers perceive that the returns to education are higher for their children. This is certainly plausible in the Chinese context, moreover, because more educated parents generally have better access to nonfarm jobs, that is, jobs that have higher educational requirements and offer higher incomes than farming (e.g., Dong and Bowles 2002; Zhang, Huang, and Rozelle 2002; Zhang and Li 2003).

Second, mothers were asked how many years of schooling they wished their children could achieve in the absence of any constraints on finances, ability, or time. Some 61.5% of women in the lowest two quintiles of mother's education responded that they wished their children could attend university. By contrast, 75.9% of women in the highest two quintiles of mother's education responded that they wished their children could attend university. This finding suggests that more educated households may have preferences for higher levels of schooling among children. Both higher returns to educated parents invest more in their children's human capital.

VI. Conclusion

The literature has documented a strong association between parental education and child human capital development, a relationship that persists despite the inclusion of controls for household and community background factors. This relationship is often attributed to higher levels of investment in children's human capital made by more educated parents, but the nature of such investments has not been well understood. Investments in human capital may include spending on educational goods and services and time spent interacting with children for educational purposes, yet parents in resource-constrained households in areas with incomplete credit markets are likely to face a tradeoff between these investments. Because more educated parents are likely to earn higher wages, the opportunity cost of time spent outside the workplace is high, and these parents may spend less time interacting with children in

order to provide more goods for children's human capital development. However, more educated parents are likely to be more adept at teaching children in the home; thus they may forgo some time in the workplace in order to provide more time interacting with children. Finally, more educated parents may provide higher levels of both investments despite being resource-constrained if the returns to children's human capital development differ for their children or if such parents have different preferences for children's education.

In this study, the determinants of six different educational investments are estimated using a survey of children, households, schools, and communities in northwestern China. It is shown that more educated parents are not substituting goods investments for time investments or vice versa but are, instead, providing higher levels of both types of investments. This result is generally robust to conditioning on a rich set of controls, including household wealth, teacher quality, child cognitive development, and community fixed effects, suggesting that the relationship between parental education and educational investments is quite strong. Evidence suggests that the perceived returns to education are higher for the children of more educated mothers. In addition, more educated mothers are shown to prefer more education for their children. Greater investment in both goods and time among more educated parents is thus likely explained by both higher expected returns and different preferences for education.

A second key finding is that the marginal effect of education differs for mothers and fathers: an additional year of mother's education leads to greater time investments than an additional year of father's education. This finding is particularly true for sons who scored below average on a cognitive development test and for daughters who scored above average. Mother's education also has a stronger impact on investments in some goods investments for lowscoring sons. An additional year of father's education raises nonrequired expenditures on sons' education and the probability that the sampled child has access to reading materials by more than mother's education, although this result is very sensitive to the empirical specification. The pronounced difference of parental education does not have a strong, systematic gender bias in investments; promoting education in time investments suggests that mothers either have a greater interest in their children's schooling or that they have a lower opportunity cost of time.

Finally, evidence of parental bias in allocating educational investments is limited. Specifically, fathers weakly favor sons and mothers weakly favor daughters in some educational investments. Nevertheless, these results are sensitive to the specification.

Therefore, if the relationship between parental education and investments

in children's human capital persists for future generations, then promoting educational attainment among the current generation of students will increase educational investments for their children, leading to higher levels of schooling and greater investments in their children—a virtuous cycle. Furthermore, because an additional year of mother's education generally increases investment in children's human capital by more than an additional year of father's education, and because mother's education does not have a strong, systematic gender bias in investments, promoting education in the current generation of girls is likely to have enduring effects for future generations of both girls and boys in rural China.

References

- Behrman, Jere R., Andrew D. Foster, Mark D. Rosenzweig, and Prem Vashishta. 1999. "Women's Schooling, Home Teaching, and Economic Growth." *Journal of Political Economy* 107, no. 4:682–714.
- Behrman, Jere R., Shahrukh Khan, David Ross, and Richard Sabot. 1997. "School Quality and Cognitive Achievement Production: A Case Study for Rural Pakistan." *Economics of Education Review* 16, no. 2:127–42.
- Biemer, Paul P., Robert M. Groves, Lars E. Lyberg, Nancy A. Mathiowetz, and Seymour Sudman, eds. 2004. *Measurement Errors in Surveys*. New York: Wiley-Interscience.
- Brown, Philip H., and Albert Park. 2002. "Education and Poverty in Rural China." *Economics of Education Review* 21, no. 6:523-41.
- Case, Anne, and Angus Deaton. 1999. "School Inputs and Educational Outcomes in South Africa." *Quarterly Journal of Economics* 114, no. 3:1047-87.
- Coleman, James S., Ernest Q. Campbell, Carol J. Hobson, James McPartland, Alexander M. Mood, Frederick D. Weinfeld, and Robert L. York. 1966. *Equality of Educational Opportunity*. Washington, DC: Office of Education, U.S. Department of Health, Education, and Welfare.
- Connelly, Rachel, and Zhenzhen Zheng. 2003. "Determinants of School Enrollment and Completion of 10 to 18 Year Olds in China." *Economics of Education Review* 22, no. 4:379–88.
- Davis, Deborah. 1992. "Job Mobility in Post-Mao Cities: Increases on the Margin." China Quarterly 119:577–97.
- Dong, Xiaoyuan, and Paul Bowles. 2002. "Segmentation and Discrimination in China's Emerging Industrial Labor Market." *China Economic Review* 13, nos. 2–3: 170–96.
- Du, Yang, Albert Park, and Sangui Wang. 2005. "Migration and Rural Poverty in China." *Journal of Comparative Economics* 33, no. 4:688–709.
- Freeman, Howard E., Robert E. Klein, John W. Townsend, and Aaron Lechtig. 1980. "Nutrition and Cognitive Development among Rural Guatemalan Children." *American Journal of Public Health* 70, no. 12:1277–85.
- Glewwe, Paul, and Hanan Jacoby. 1994. "Student Achievement and Schooling Choice

in Low Income Countries: Evidence from Ghana." Journal of Human Resources 29, no. 3:841-64.

- Hannum, Emily. 1998. "Educational Inequality: Hidden Consequences of the Reform Era in Rural China." PhD thesis, University of Michigan.
- ——. 2005. "Market Transition, Educational Disparities, and Family Strategies in Rural China: New Evidence on Gender Stratification and Development." *Demography* 42, no. 2:275–99.
- Hannum, Emily, and Yu Xie. 1994. "Trends in Educational Gender Inequality in China, 1949–1985." *Research in Social Stratification and Mobility* 13:73–98.
- Hanushek, Eric. 1997. "Assessing the Effects of School Resources on Student Performance: An Update." *Educational Evaluation and Policy Analysis* 19, no. 2:141–64.
- Hofferth, Sandra L. 1999. "Family Reading to Young Children: Social Desirability and Cultural Biases in Reporting." Unpublished manuscript, Institute for Social Research, University of Michigan.
- Hofferth, Sandra L., and Kermyt G. Anderson. 2003. "Are All Dads Equal? Biology versus Marriage as a Basis for Paternal Investment." *Journal of Marriage and Family* 65:213–32.
- Juster, F. Thomas. 1985. "Conceptual and Methodological Issues Involved in the Measurement of Time Use." In *Time, Goods, and Well-Being*, ed. F. Thomas Juster and Frank P. Stafford. Ann Arbor, MI: Institute for Social Research.
- Knight, John, and Lina Song. 1993. "Why Urban Wages Differ in China." In The Distribution of Income in China, ed. Keith Griffin and Zhao Renwei, 217–84. London: Macmillan.
- Kong, Peggy. 2003. "The Nature of Parental Involvement in Children's Schooling in a Poor Rural Province in Northwestern China." Unpublished manuscript, Graduate School of Education, Harvard University.
- Li, Hongbin, and Junsen Zhang. 2004. "Testing the External Effect of Household Behavior: The Case of the Demand for Children." Unpublished manuscript, Department of Economics, Chinese University of Hong Kong.
- Lillard, Lee A., and Robert J. Willis. 1994. "Intergenerational Educational Mobility: Effects of Family and State in Malaysia." *Journal of Human Resources* 29, no. 4: 1126–66.
- Park, Albert, and Emily Hannum. 2001. "Do Teachers Affect Learning in Developing Countries? Evidence from Matched Student-Teacher Data from China." Paper presented at the Rethinking Social Science Research on the Developing World in the 21st Century Conference, Park City, UT, June 7–10.
- Robinson, John P. 1985. "The Validity and Reliability of Diaries versus Alternative Time Use Measures." In *Time, Goods, and Well-Being*, ed. F. Thomas Juster and Frank P. Stafford. Ann Arbor, MI: Institute for Social Research.
- Sargent, Tanja, and Emily Hannum. 2005. "Keeping Teachers Happy: Job Satisfaction among Primary School Teachers in Rural Northwest China." *Comparative Education Review* 49, no. 2:173–204.
- Sathar, Zeba A., and Cynthia B. Lloyd. 1994. "Who Gets Primary Schooling in Pakistan? Inequalities among and within Families." *Pakistan Development Review* 33, no. 2:103–34.

- Solinger, Dorothy. 1999. Contesting Citizenship in Urban China: Peasants, Migrants, the State, and the Logic of the Market. Berkeley: University of California Press.
- Strauss, John, and Duncan Thomas. 1995. "Human Resources: Empirical Modeling of Household and Family Decisions." In *Handbook of Development Economics*, vol. 3, ed. J. Behrman and T. N. Srinivasan, 1883–2023. Amsterdam: Elsevier Science.
- Thomas, Duncan. 1994. "Like Father Like Son, or, Like Mother Like Daughter: Parental Education and Child Health." *Journal of Human Resources* 29, no. 4: 950–88.
- Tsang, Mun C. 1994. "Costs of Education in China: Issues of Resource Mobilization, Equality, Equity, and Efficiency." *Education Economics* 2, no. 3:287–312.
- Unger, Jonathan. 2002. The Transformation of Rural China. Armonk, NY: M. E. Sharpe.
- Wang, Chengzhi. 2002. "Minban Education: The Planned Elimination of the People-Managed Teachers in Reforming China." International Journal of Educational Development 22, no. 2:109–39.
- Wu, Xiaogang, and Donald J. Treiman. 2004. "The Household Registration System and Social Stratification in China, 1955–1996." *Demography* 41, no. 2:363–84.
- Zellner, Arnold. 1962. "An Efficient Method of Estimating Seemingly Unrelated Regressions and Tests for Aggregation Bias." *Journal of the American Statistical Association* 57:348–68.
- Zhang, Linxiu, Jikun Huang, and Scott Rozelle. 2002. "Employment, Emerging Labor Markets, and the Role of Education in Rural China." *China Economic Review* 13, nos. 2–3:313–28.
- Zhang, Xiaobo, and Guo Li. 2003. "Does *Guanxi* Matter to Nonfarm Employment?" *Journal of Comparative Economics* 31, no. 2:315–31.